

Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2023GL106689

Key Points:

- Shallow and small waterbodies have high but variable rates of gross primary production and respiration
- Depth plays a large role in metabolism rates
- Shallow waterbodies are some of the most productive freshwater ecosystems, with implications for carbon cycling and ecosystem dynamics

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

J. S. Rabaey, rabae005@umn.edu

Citation:

Rabaey, J. S., Holgerson, M. A., Richardson, D. C., Andersen, M. R., Bansal, S., Bortolotti, L. E., et al. (2024). Freshwater biogeochemical hotspots: High primary production and ecosystem respiration in shallow waterbodies. *Geophysical Research Letters*, *51*, e2023GL106689. https://doi.org/10.1029/ 2023GL106689

Received 9 NOV 2023 Accepted 22 JUL 2024

Author Contributions:

Conceptualization: Joseph S. Rabaey, Meredith A. Holgerson, David C. Richardson Data curation: Joseph S. Rabaey, Meredith A. Holgerson, David C. Richardson Formal analysis: Joseph S. Rabaey, Meredith A. Holgerson, David C. Richardson, Mikkel R. Andersen, Sheel Bansal, Lauren E. Bortolotti, James B. Cotner, Daniel J. Hornbach, Kenneth T. Martinsen, Eric K. Moody, Olivia F. Schloegel Investigation: Joseph S. Rabaey, Meredith A. Holgerson, David

© 2024. The Author(s).

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Freshwater Biogeochemical Hotspots: High Primary Production and Ecosystem Respiration in Shallow Waterbodies

Joseph S. Rabaey¹, Meredith A. Holgerson², David C. Richardson³, Mikkel R. Andersen⁴, Sheel Bansal⁵, Lauren E. Bortolotti⁶, James B. Cotner^{1,7}, D, Daniel J. Hornbach⁸, Kenneth T. Martinsen⁹, Eric K. Moody¹⁰, and Olivia F. Schloegel¹¹

¹Department of Ecology, Evolution, and Behavior, University of Minnesota – Twin Cities, St. Paul, MN, USA, ²Department of Ecology and Evolutionary Biology, Cornell University, Ithaca, NY, USA, ³Biology Department, State University of New York at New Paltz, New Paltz, NY, USA, ⁴Centre for Freshwater and Environmental Studies, Dundalk Institute of Technology, Dundalk, Ireland, ⁵U.S. Geological Survey, Northern Prairie Wildlife Research Center, Jamestown, ND, USA, ⁶Institute for Wetland and Waterfowl Research, Ducks Unlimited Canada, Stonewall, MB, Canada, ⁷Department of Biological Sciences, University of Bergen, Bergen, Norway, ⁸Department of Environmental Studies, Macalester College, St. Paul, MN, USA, ⁹Freshwater Biological Laboratory, Department of Biology, University of Copenhagen, Copenhagen, Denmark, ¹⁰Department of Biology, Middlebury College, Middlebury, VT, USA, ¹¹Department of Biological Sciences, Kent State University, Kent, OH, USA

Abstract Ponds, wetlands, and shallow lakes (collectively "shallow waterbodies") are among the most biogeochemically active freshwater ecosystems. Measurements of gross primary production (GPP), respiration (R), and net ecosystem production (NEP) are rare in shallow waterbodies compared to larger and deeper lakes, which can bias our understanding of lentic ecosystem processes. In this study, we calculated GPP, R, and NEP in 26 small, shallow waterbodies across temperate North America and Europe. We observed high rates of GPP (mean 8.4 g $O_2 m^{-3} d^{-1}$) and R (mean $-9.1 g O_2 m^{-3} d^{-1}$), while NEP varied from net heterotrophic to autotrophic. Metabolism rates were affected by depth and aquatic vegetation cover, and the shallowest waterbodies had the highest GPP, R, and the most variable NEP. The shallow waterbodies from this study had considerably higher metabolism rates compared to deeper lakes, stressing the importance of these systems as highly productive biogeochemical hotspots.

Plain Language Summary Ecosystem metabolism is an important metric for understanding how an ecosystem uses and transforms energy and materials. In aquatic ecosystems, ecosystem metabolism is most often calculated by measuring oxygen. Oxygen is produced by aquatic plants and algae through photosynthesis, and oxygen is consumed by bacteria and other organisms through aerobic respiration. Oxygen production and consumption (metabolism) indicate how organic matter is flowing through the ecosystem, and whether an aquatic system is predominantly respiring (indicating it is a carbon source), or photosynthesizing (indicating it is a carbon sink). Metabolism rates in freshwater lakes scale with surface area and depth, but there are few metabolism rates calculated from oxygen measurements in shallow waterbodies. Ponds, wetlands, and shallow lakes are the most abundant freshwater systems on Earth and may function differently from larger lakes. In this study, we show that shallow systems are incredibly productive, with some of the highest rates of metabolism among aquatic ecosystems, and that depth is one of the most important factors affecting metabolism rates.

1. Introduction

Ponds, shallow lakes, and wetlands (collectively "shallow waterbodies") are biogeochemical hotspots with high variability (Bansal et al., 2023; Cheng & Basu, 2017; Downing, 2010; Holgerson & Raymond, 2016). To better understand why shallow waterbodies have high rates of biogeochemical activities, we need large-scale, multi-site studies to assess how ecosystem functions scale across size and depth gradients (e.g., Cheng et al., 2023; Hansen et al., 2018; Qin et al., 2020; Richardson et al., 2022). An important measure of ecosystem function is ecosystem metabolism, an integrative measure of aquatic primary production and aerobic respiration of organic carbon. Ecosystem metabolism includes rates of gross primary production (GPP), ecosystem respiration (R), and net ecosystem production (NEP), and describes whether waterbodies are ultimately autotrophic (GPP > R) or heterotrophic (GPP < R).





C. Richardson, Mikkel R. Andersen, Sheel Bansal, Lauren E. Bortolotti, James B. Cotner, Daniel J. Hornbach, Kenneth

T. Martinsen, Eric K. Moody, Olivia

F. Schloegel

Writing – original draft: Joseph

S. Rabaey, Meredith A. Holgerson, David C. Richardson

Writing – review & editing: Mikkel

R. Andersen, Sheel Bansal, Lauren

E. Bortolotti, James B. Cotner, Daniel

J. Hornbach, Kenneth T. Martinsen, Eric

K. Moody, Olivia F. Schloegel

Ecosystem metabolism in lakes varies due to both waterbody size and depth, indicating that understudied shallow waterbodies may have different functional responses to environmental drivers than well-studied larger lakes (Brylinsky & Mann, 1973; Staehr et al., 2012). For example, volumetric GPP typically decreases with increasing lake size and depth (Duarte & Kalff, 1989), yet there can also be high variability among both small and shallow waterbodies (Hornbach et al., 2017, 2020; Martinsen et al., 2017; Richardson et al., 2022; Staehr et al., 2012). Similarly, most shallow waterbodies have high, but variable R, often leading to net heterotrophic conditions (Hagerthey et al., 2010; Hoellein et al., 2013). Heterotrophy is likely driven by increased coupling between the sediments and water column as well as a strong edge effect, which increases terrestrial carbon entering these waterbodies (Holgerson, 2015; Rubbo et al., 2006). Yet, the low number of metabolism studies in small or shallow waterbodies a more comprehensive understanding of freshwater metabolism.

Water depth may mechanistically change metabolism in shallow waterbodies. Specifically, benthic processes can affect the entire water column in shallower systems, and therefore lead to higher rates of open-water R (Staehr et al., 2012). Similarly, shallow depths promote growth of submerged and emergent plants, which contribute to metabolism through photosynthesis, providing labile carbon for respiration, providing physical structures for epiphytes, and altering the light environment (Bortolotti et al., 2019; Burkholder & Wetzel, 1989; Hornbach et al., 2020; Martinsen et al., 2017; Rabaey et al., 2021). Macrophyte production can occur either inside or outside of the water column depending on plant growth forms (e.g., submerged vs. emergent), thus affecting the measurement of GPP (Hagerthey et al., 2010; Sand-Jensen et al., 2019). Shallow waterbodies are also prone to water column anoxia when R greatly exceeds GPP, for example, due to terrestrial inputs of organic carbon or water column drawdown (Martinsen et al., 2019), or when macrophytes shade the lower parts of the water column (Andersen, Kragh, & Sand-Jensen, 2017; Rabaey & Cotner, 2022).

In this study, we aimed to (a) quantify ecosystem metabolism (GPP, R, and NEP) across a range of shallow waterbodies, (b) partition background R from R closely tied to GPP (c) identify drivers of GPP, R, and NEP in these systems, and (d) assess how metabolism differs across shallow waterbody types (i.e., ponds, wetlands, shallow lakes). We predicted that shallow waterbodies are net heterotrophic with higher rates of GPP and R than most aquatic systems. Further, we expected that metabolism rates would be most affected by water column depth, waterbody surface area, and macrophyte coverage and type.

2. Methods

2.1. Field Sampling

We sampled 35 freshwater shallow (maximum depth < 4.8 m) and relatively small (surface area < 17 ha) waterbodies, mostly located across temperate North America with one site in Europe (Figure S1 in Supporting Information S1). All waterbodies fell under definitions of ponds (maximum depth < 5 m, surface area < 5 ha), shallow lakes (<5 m, >5 ha), or wetlands (<5 m, >30% emergent vegetation cover) described in Richardson et al. (2022). The wetlands in this study all had an open water location where sensors were placed, surrounded by vegetation that comprised over 30% of the basin. Sampling took place during the open water period between 2010 and 2022, with only one season of data used for each waterbody. To calculate ecosystem metabolism, we measured near-surface dissolved oxygen (DO) and water temperature using high-frequency sensors (Table S1 in Supporting Information S1). Photosynthetically active radiation (PAR) was measured either on site (16 waterbodies) or at nearby meteorological stations. Physical, chemical, and biological characteristics (hereafter "waterbody characteristics") were measured at least once during the study period (Table S2 in Supporting Information S1).

2.2. Metabolism Modeling

To estimate volumetric GPP, R, and NEP, we fit ecosystem metabolism models using the DO, PAR, and temperature data with the maximum likelihood method in the *LakeMetabolizer* R package (Winslow et al., 2016). We used a simple metabolism model that assumed a fully mixed water column and assumed a constant k_{600} value to estimate the gas exchange coefficient (*k*). We used a constant k_{600} value of 0.654 m day⁻¹, as this was the average k_{600} estimated for 30 small waterbodies (surface area < 10 ha) via floating chambers or gas tracers (Holgerson et al., 2017).

We compared this simple model to three more complex models: one that used wind to estimate k (Vachon & Prairie, 2013), one that used water column thermistors to calculate the mixing depth (z_{mix}), and one that used both wind-derived k and temperature-derived z_{mix} . Even though shallow systems can often stratify intermittently (Andersen, Sand-Jensen, et al., 2017; Holgerson, Richardson, et al., 2022) and k_{600} values can vary across systems (Holgerson et al., 2017), the simple model was similar to the more complex models in both goodness of fit and estimated metabolism values (Text S1, Table S3, Figures S2 and S3 in Supporting Information S1). Therefore, all further analyses used the simplest model to maximize the number of included waterbodies as only a subset had onsite wind data required to estimate k (19 waterbodies) or thermistor data required to estimate z_{mix} (12 waterbodies).

2.3. Drivers of Model Fit

For each daily model fit on each waterbody, we visually inspected the observed and modeled DO curves along with DO residuals, PAR, and fit statistics (e.g., residual sums of squares). Following Richardson et al. (2017), each daily fit was inspected by at least three team members independently to determine if the model fit was "good" or "bad" (see Text S1–S3; Figure S4 in Supporting Information S1), with impossible values (R > 0, GPP < 0) automatically deemed as bad model fits.

We then assessed relationships between waterbody characteristics and model fit at the daily scale, including daily Δ DO (difference between DO at sunrise and DO at sunset), daily duration of anoxia (<1 mg L⁻¹ DO), mean daily PAR, and mean daily temperature, as well as waterbody characteristics with the proportion of good model fit days for each of the 35 waterbodies. For all comparisons, we only included data from the summer period (21 June–22 September) to reduce seasonality among waterbodies with different sampling periods, and all waterbodies were required to have >14 days of data (regardless of good or bad model fits) within this period. To relate daily model fits (good vs. bad) across all waterbodies to possible drivers, we used a binomial generalized linear mixed-effects model with a logit link function using the *lme4* package in R (Bates et al., 2015). To compare site-level drivers, the proportion of good model fit days was regressed against waterbody characteristics using a linear model or a log-linear model, with the best model being chosen by Akaike Information Criterion for small sample sizes (AICc).

After testing drivers of model fit, we further split the data to only include waterbodies with >14 days of good model fits within the summer period. This resulted in 26 of the total 35 initially eligible waterbodies, which we used for the remainder of analyses.

2.4. Calculating Background Metabolism

We fit regression models to describe the relationship between volumetric rates of R and GPP in each waterbody to estimate coupling between GPP and R (β_1) and calculate background respiration ($R_{background}$; β_0) (Text S2 in Supporting Information S1). $R_{background}$ represents the respiration of less-labile organic material that decomposes slowly and is more stable through time, including allochthonous (externally derived) organic matter, recalcitrant autochthonous organic matter, and in situ senescence of macrophytes (Solomon et al., 2013). We compared the $R_{background}$ with coupling parameters using semi-log transformed regression. We determined if differences in $R_{background}$ across waterbodies related to waterbody characteristics using a classification and regression tree (CART) analysis using the rpart package in R (Therneau & Atkinson, 2022).

2.5. Drivers of Metabolism and Waterbody Comparisons

To assess daily drivers of metabolism, we used a linear mixed-effects model to regress daily rates against daily variables using the *lme4* package in R (Bates et al., 2015). For site-level drivers, we regressed mean rates of GPP, R, and NEP (waterbody mean) against waterbody characteristics. As GPP cannot be negative, and R cannot be positive, we compared a simple linear model and an exponential model (asymptote at 0) for GPP and R versus waterbody characteristics and chose the best model based on AICc. For NEP, we compared a linear model as well as a second-order polynomial model.

To put our results in context, we compared our data to three previous syntheses of aquatic ecosystem metabolism: Hoellein et al. (2013), Holgerson, Hovel, et al. (2022), and Richardson et al. (2022). We removed repeated data across the three data sets and used waterbody surface area and maximum depth to classify waterbodies as in Richardson et al. (2022) as ponds, shallow lakes, small lakes (>5 m, <5 ha), lakes (>5 m, >5 ha), and wetlands.



Geophysical Research Letters



Figure 1. Proportion of good metabolism model fit days per shallow waterbody (Fit proportion) versus (a) mean daily change in dissolved oxygen (Δ DO) from sunrise to sunset ($R^2 = 0.39$, p < 0.001), (b) mean DO ($R^2 = 0.44$, p < 0.001), (c) dissolved organic carbon (DOC) concentrations ($R^2 = 0.57$, p < 0.001), and (d) total dissolved nitrogen (TDN) concentrations ($R^2 = 0.52$, p < 0.001). Circles represent the 26 waterbodies that were used in subsequent analyses, triangles represent the 9 waterbodies that did not have enough good metabolism fits to analyze further. Model parameter estimates are given in Table S4 in Supporting Information S1.

To make our results comparable, we scaled our volumetric rates to areal rates using 66% of the maximum depth for each waterbody which was the mean mixing depth of waterbodies where mixing data were available (n = 12). We then compared areal GPP, R, and NEP across these five aquatic waterbody types using non-parametric Kruskal-Wallis tests with Wilcoxon post-hoc comparisons (Hochberg p-value adjustment) if at least one median was different. For all analyses, we used R statistical software (R 4.4.2, R Core Team 2022).

3. Results

3.1. Drivers of Model Fit

The proportion of good model fits for the 35 waterbodies ranged from 4% to 100% of days (mean 59%). Within waterbodies, the probability that each day was a good or bad model fit was related to the daily duration of anoxia and the daily ΔDO (Figure S5 in Supporting Information S1). Across waterbodies, higher proportions of good fit days were related to higher mean daily ΔDO , mean DO concentrations, and dissolved organic carbon (DOC) and total dissolved nitrogen (TDN) concentrations (Figure 1).

3.2. Drivers of GPP, R, and NEP

Mean volumetric rates (g $O_2 m^{-3} d^{-1}$) averaged by waterbody ranged by an order of magnitude for GPP (1.1–24.0; mean 8.4), R (-1.3 to -28.0; mean -9.1), and NEP (-5.0 to 3.6; mean -0.6). Daily rates were as high as 38 g $O_2 m^{-3} d^{-1}$ for GPP and -43 g $O_2 m^{-3} d^{-1}$ for R (Figure 2). Mean NEP was negative in 15 waterbodies and



Figure 2. Daily volumetric gross primary production (GPP; gray circles) and respiration (R; black triangles) for the 26 shallow waterbodies. Panels are colored by waterbody type, ponds (yellow), wetlands (green), and shallow lakes (blue). Note the *y*-axis differs across waterbodies. Gaps in the data are times when data was unavailable, or when metabolism values were removed due to poor model fits. Refer to Table S2 in Supporting Information S1 for site descriptions.

positive in 11, though for 9 waterbodies with a positive mean NEP and 2 waterbodies with negative mean NEP the 95% confidence interval included 0.

When relating mean waterbody metabolism to waterbody characteristics, the magnitude of GPP and R exponentially decreased with increasing maximum depth, especially for waterbodies <2 m deep (Figures 3a and 3d). GPP and R did not relate to waterbody surface area, whereas NEP significantly increased with increasing surface area (Figure 3h). Both the most positive and most negative NEP were observed in the shallowest waterbodies, with less variability in NEP amongst larger (>5 ha) and deeper (>2 m) waterbodies. The magnitude of GPP and R increased with increasing emergent vegetation cover (Figures 3c and 3f), though this was driven by one wetland with high (80%) emergent cover. NEP had a unimodal relationship with emergent cover, with intermediate amounts of emergent cover equating to a positive NEP, while NEP was negative where emergent vegetation was



Geophysical Research Letters



Figure 3. Relationships of volumetric gross primary production (GPP), respiration (R), and net ecosystem production (NEP) with maximum depth, surface area, and emergent vegetation cover in shallow waterbodies. Each point is the mean rate, error bars represent ±1 SE. Trend lines represent significant relationships, including (a) GPP versus Max depth ($R^2 = 0.51$, p < 0.001), (c) GPP versus emergent vegetation cover ($R^2 = 0.23$, p = 0.029), (d) R versus Max depth ($R^2 = 0.68$, p < 0.001), (f) R versus emergent vegetation cover ($R^2 = 0.34$, p < 0.0053), (h) NEP versus surface area ($R^2 = 0.15$, p = 0.050), (i) NEP versus emergent vegetation cover ($R^2 = 0.37$, p < 0.016). Twenty-one waterbodies had data for emergent cover (c, f, i), and only one waterbody had high emergent cover (over 40%). With this 80% emergent cover point removed, relationships in (c) and (f) are no longer significant, while (i) remains significant. Model parameter estimates are given in Table S4 in Supporting Information S1.

either low or very high. No other waterbody characteristics (e.g., nutrients, chl *a*) were significantly correlated with mean waterbody GPP, R, and NEP (Figure S6 in Supporting Information S1). Daily rates of NEP were most strongly correlated with the mean daily DO concentration and the density gradient across the water column (Figure S7 in Supporting Information S1).

3.3. Background Respiration Drivers

Coupling of R to GPP was strong in more than half of the waterbodies (15 out of 26), indicated by coupling not being significantly different than -1 (-1 indicating perfect R-GPP coupling, Figure S9 in Supporting Information S1). R_{background} was <0 in the majority of systems (15 out of 26, Figure S10 in Supporting Information S1), indicating the importance of allochthonous or less-labile organic matter metabolism. R_{background} was minimal (less negative) where R-GPP coupling was most negative (Figure 4a, R_{background} = $-e^{4.0+4.7*\beta_1} + 0.21$, p < 0.001, $F_{1,22} = 14.7$), suggesting that autochthonous labile carbon production exceeded allochthonous contributions in those waterbodies. Maximum depth was the only predictor of R_{background} from the pruned regression tree





Figure 4. Background respiration ($R_{background}$) related to (a) the coupling between respiration (R) and gross primary production (GPP) where a value closer to -1 indicates strong coupling of respiration and photosynthesis and (b) maximum depth of the waterbody, where a regression tree identified a split (vertical dashed line) with higher $R_{background}$ in waterbodies shallower than 1.5 m maximum depth.

(Figure 4b) with a split at 1.5 m maximum depth. Median $R_{background}$ was $-5.6 \text{ g O}_2 \text{ m}^{-3} \text{ d}^{-1}$ in sites with <1.5 m maximum depth and $-1.5 \text{ g O}_2 \text{ m}^{-3} \text{ d}^{-1}$ sites with >1.5 m maximum depth.

3.4. Comparing Ecosystem Metabolism Across Waterbody Types

When aggregated with areal metabolism rates reviewed in Hoellein et al. (2013), the waterbodies from this study have among the highest metabolism rates measured in freshwater ecosystems (Figure 5a). When excluding streams and adding additional published lentic metabolism rates, we found that ponds and wetlands were on the highest end of areal metabolism measured relative to lakes (Figure 5b). Small lakes had significantly lower areal GPP (Kruskal-Wallis: $\chi^2 = 30.7$, df = 4, p < 0.001) and R ($\chi^2 = 21.0$, df = 4, p < 0.001) than all other waterbodies (ponds, wetlands, shallow lakes, and lakes). Areal NEP differed among the five waterbody types (Figure 5c, Kruskal-Wallis: $\chi^2 = 25.4$, df = 4, p < 0.001). Lakes had the highest NEP and were balanced between GPP and R;



Figure 5. (a) Areal gross primary production (GPP) and respiration (R) from this study (filled circles), separated out by pond, wetland, and shallow lake overlaid with a recreated Hoellein et al. (2013) Fig. 1c in greyscale. (b) All pond and wetland metabolism rates from this study and literature (filled circles) overlaid over the extent of all lakes, small lakes, and shallow lakes from both this study and published values including reclassifying ecosystem types from Hoellein et al. (2013). Note the log scale GPP and R axes in panels (a) and (b); R is presented as positive to match Hoellein et al. (2013). (c) Areal net ecosystem production (NEP; g $O_2 m^{-2} d^{-1}$) from this study and the literature across five waterbody types: Pond (mean ± sd = -1.6 ± 5.1), Wetland (Wetl.) (-2.6 ± 2.1), shallow lake (Shall.) (-0.4 ± 3.9), small lake (Small) (-1.3 ± 2.6), and Lake (0.4 ± 3.5). One shallow lake (NEP = -18.5) was excluded from the graph for ease of viewing but was included in the statistical analysis. Letters in panel (c) indicate groups of similar NEP values from post-hoc pairwise Wilcoxon tests. Box plots depict the minimum, first quartile, median, third quartile, and maximum, overlain with each individual point.

but there were both autotrophic and heterotrophic waterbodies (Figure 5c). The most heterotrophic waterbodies were wetlands, followed by ponds and small lakes (Figure 5c).

4. Discussion

We found that open-water metabolism rates corresponded strongly to waterbody size and maximum depth with ponds, shallow lakes, and wetlands differing in ecosystem metabolism compared to larger lakes (Figure 5). Specifically, the shallowest waterbodies had the highest volumetric rates of GPP and R, and were the most variable in terms of NEP; we also found that net heterotrophy increased with decreasing surface area (Figure 3). Overall, the rates of GPP and R of the 26 waterbodies in our study are some of the highest reported among freshwater ecosystems.

4.1. Drivers of Metabolism in Shallow Waterbodies

While depth and surface area have been strongly correlated in previous studies of lake metabolism (Staehr et al., 2012), depth and area were not correlated in our study (R = 0.22, p = 0.21), allowing us to disentangle their individual effects. Despite all waterbodies being <5 m deep, depth strongly predicted ecosystem metabolism and R_{background}; however, surface area only predicted NEP. Larger waterbodies had more positive NEP, possibly due to a smaller edge effect that could reduce shading from forest canopy and reduce allochthonous nutrients and DOC inputs per water volume. In contrast to surface area, depth directly predicted R and GPP. The shallowest systems (<1.5 m maximum depth) had the greatest R, R_{background}, and R-GPP decoupling, indicating a greater role of allochthonous or more recalcitrant organic matter similar to lakes (e.g., del Giorgio & Williams, 2005; Solomon et al., 2013). Benthic-pelagic coupling is stronger in shallow waterbodies, which allows for sediment respiration to have a stronger effect on open-water metabolism (Richardson et al., 2022; Staehr et al., 2012) and R_{background}. The importance of stronger stratification had higher NEP (Figure S7 in Supporting Information S1) as stratification would weaken the impact that sediment R has on epilimnetic metabolism.

Depth also predicted GPP, which is likely because shallow waterbodies can have light penetration throughout the entire water column, allowing GPP by both phytoplankton and submerged macrophytes. With adequate light, nutrient-rich sediments can boost GPP, perhaps explaining why our shallowest systems had some of the highest rates of open-water GPP. The magnitude and variability of open-water GPP and NEP may also be explained by aquatic macrophytes, including submerged (Brothers et al., 2013; Rabaey et al., 2021), floating (Allende & Fontanarrosa, 2023; Rabaey & Cotner, 2022), and emergent plants, which contrasts phytoplankton-driven GPP in deeper systems. Submerged macrophytes can increase GPP through photosynthesis and by providing substrata for epiphytes (Burkholder & Wetzel, 1989), and can also increase R when macrophyte beds are dense (Theus et al., 2023). While submerged macrophyte production and respiration affect DO, emergent macrophytes have limited direct impact on water-column GPP with photosynthesis occurring above the water-air interface. However, they can contribute to both open-water and sediment R by releasing carbon and fueling heterotrophic decomposition (Barko et al., 1991). We found that waterbodies with higher emergent vegetation coverage (i.e., freshwater wetlands) had greater R and lower NEP, though largely driven by one waterbody with 80% emergent coverage. As wetlands are highly variable in hydrology, vegetation, and soil properties, more research should explore how different plant communities influence ecosystem metabolism. Furthermore, our research suggests that traditional indicators of trophic status (e.g., nutrients, chl a, secchi depth) which often predict GPP in lakes (e.g., Hoellein et al., 2013) may not apply for shallow waterbodies (Figure S6 in Supporting Information S1).

4.2. Using Open-Water Metabolism Models in Small Waterbodies

Using the open-water metabolism technique (Hanson et al., 2003; Odum, 1956) in shallow waterbodies captures an integrated measure of pelagic and benthic metabolism, but comes with limitations. Specifically, we were unable to model metabolism for 48% of days across our 35 initial waterbodies (Table S3 in Supporting Information S1), a challenge consistent with other studies of shallow waterbody metabolism (Martinsen et al., 2019). Poor model fits were related to hypoxia or limited diel DO changes (Figure 1, Figure S5 in Supporting Information S1) as many waterbodies went anoxic for days or weeks during the summer period. Anoxia does not necessarily signify low productivity; rather high respiration rates can mask high productivity rates in shallow waters (e.g., Holgerson, Post, & Skelly, 2016), though this may still bias our results by removing days with large

19448007, 2024, 15, Downloaded from https:/ rary.wiley.com/doi/10.1029/2023GL106689 by Royal Danish Library, Wiley Online Library on [10/08/2024]. See the Terms and Conditions (https://www.actional.com/doi/10.1029/2023GL106689 by Royal Danish Library, Wiley Online Library on [10/08/2024]. Wiley Online Library for rules use; OA articles are governed by the applicable Creative Cc

Acknowledgments

This work benefited from participation in or use of the Global Lake Ecological Observatory Network (GLEON). We thank all municipalities and landowners who granted us access to study sites. including the Ducks Unlimited Canada (DUC) Yorkton Office, Katharine Ordway Natural History Study Area, Warner Nature Center, and Mary Cullen. We are grateful to the following people for assistance during field work: I. Potts, Brian Herwig, Mark Hanson, Will Hobbs, Leah Domine, Kyle Zimmer, Kathryn Hoffman, Margot Groskreutz, Alonso Delgado, Keith Douville, Mark Hove, Mira Ensley-Field, Matthew Glasenapp, Ian Goodbar, J. Douglas Harman, Benjamin Huber, Emily Kangas, Kira Liu, Molly Stark-Ragsdale, Long Tran, Emily Schilling, Holly Kundel, the Washington Department of Fish and Wildlife field crews, and the 2021 Aquatic Ecology class at Middlebury College. JSR was supported by the University of Minnesota. MAH was supported by the St. Olaf College Collaborative Undergraduate Research and Inquiry program and the National Science Foundation (Grant 2143449). MRA fieldwork was supported as part of the BEYOND 2020 project (grant-aid agreement no. PBA/FS/16/02) and by the Marine Institute, as well as funded under the Marine Research Programme by the Irish Government. SB was partially funded by the Genome Sciences Program of the United States Department of Energy, Office of Biological and Environmental Research (Grant DE-SC0023084). DJH fieldwork was supported by Macalester College and Augsburg University. JBC fieldwork was supported by the National Science Foundation (Grant DEB-0919095). LEB fieldwork was supported by Ducks Unlimited Canada's Institute for Wetland and Waterfowl Research. EKM fieldwork was supported by Middlebury College. SB and OFS fieldwork was supported by the U.S. Geological Survey, Ecosystem Mission Area, Land Change Science Climate Research and Development Program. Any use of trade, firm, or product names is for descriptive purposes only and does not imply endorsement by the U.S. Government

respiration rates. When biologically-driven DO changes are small, physical processes (i.e., wind, nighttime mixing, groundwater inputs) can have a larger relative effect that cannot be accounted for by simplified models of gas exchange, a phenomenon seen in ponds (Holgerson, Zappa, & Raymond, 2016) and oligotrophic lakes (Brothers et al., 2017; Richardson et al., 2017). For example, we observed that waterbodies undersaturated in DO frequently exhibited nighttime or early morning reaeration, likely driven by convective cooling increasing turbulence and DO exchange (Holgerson, Zappa, & Raymond, 2016; Figure S4b in Supporting Information S1). Overall, we need to better integrate variable oxygen regimes into metabolism models for shallow waterbodies, which may require improved field and modeling methods, such as eddy covariance (e.g., Berg et al., 2022), using bottle assays to artificially add DO during anoxic conditions, or approaches that monitor both DO and dissolved CO_2 (e.g., Vachon et al., 2020).

4.3. Conclusion

Small, shallow waterbodies are among the most productive of all freshwater ecosystems (Wetzel, 2001; Figure 5). Specifically, shallow systems appear to be functionally distinct from larger, deeper lakes (Richardson et al., 2022) due to shallow depths promoting strong benthic-pelagic coupling, greater heterotrophy, and greater diversity of aquatic primary producers. These high metabolic rates in shallow systems are likely significant on a global scale. For instance, if we assume that small waterbodies <10 ha have an average depth of 2.6 m and global volume of 1,780 km³ (13% of all lakes by surface area, Cael et al., 2017) and that half of annual GPP and R occurs during the summer (Text S3 in Supporting Information S1), we estimate that these small and shallow waterbodies have global rates of 1.3 Pg yr⁻¹ GPP and 1.4 Pg yr⁻¹ R, which is more than half of estimates for global rates of GPP and R from all freshwater lentic and lotic ecosystems (Maranger et al., 2018). Therefore, our findings support that shallow waterbodies are some of the most biogeochemically active systems on the planet (Bansal et al., 2023; Cheng & Basu, 2017; Downing et al., 2008; Holgerson & Raymond, 2016; Schmadel et al., 2019; Wetzel, 2001).

Data Availability Statement

Data is available in the Environmental Data Initiative (EDI) Repository (Rabaey et al., 2024). Data collected by the U.S. Geological Survey (P8, P1) can be found at Bansal et al. (2021).

References

- Allende, L., & Fontanarrosa, M. S. (2023). Addressing the trophic status in urban ponds: An evaluation of current trophic state indexes. Austral Ecology, 48(8), 1828–1847. https://doi.org/10.1111/aec.13425
- Andersen, M. R., Kragh, T., & Sand-Jensen, K. (2017). Extreme diel dissolved oxygen and carbon cycles in shallow vegetated lakes. Proceedings of the Royal Society B: Biological Sciences, 284(1862), 20171427. https://doi.org/10.1098/rspb.2017.1427
- Andersen, M. R., Sand-Jensen, K., Iestyn Woolway, R., & Jones, I. D. (2017). Profound daily vertical stratification and mixing in a small, shallow, wind-exposed lake with submerged macrophytes. *Aquatic Sciences*, 79(2), 395–406. https://doi.org/10.1007/s00027-016-0505-0
- Bansal, S., Johnson, O. F., Meier, J. A., & Tangen, B. (2021). Dissolved oxygen concentrations, light penetration, and temperature along the water-depth profile of wetlands P1 and P8 of the Cottonwood Lake Study area in North Dakota, USA, 2019 [Dataset]. U.S. Geological Survey data release. https://doi.org/10.5066/P9Q0OMYS
- Bansal, S., Post Van Der Burg, M., Fern, R. R., Jones, J. W., Lo, R., McKenna, O. P., et al. (2023). Large increases in methane emissions expected from North America's largest wetland complex. *Science Advances*, 9(9), eade1112. https://doi.org/10.1126/sciadv.ade1112
- Barko, J. W., Gunnison, D., & Carpenter, S. R. (1991). Sediment interactions with submersed macrophyte growth and community dynamics. *Aquatic Botany*, 41(1-3), 41–65. https://doi.org/10.1016/0304-3770(91)90038-7
- Bates, D., Mächler, M., Bolker, B., & Walker, S. (2015). Fitting linear mixed-effects models using LME4. Journal of Statistical Software, 67(1). https://doi.org/10.18637/jss.v067.i01
- Berg, P., Huettel, M., Glud, R. N., Reimers, C. E., & Attard, K. M. (2022). Aquatic eddy covariance: The method and its contributions to defining oxygen and carbon fluxes in Marine environments. *Annual Review of Marine Science*, 14(1), 431–455. https://doi.org/10.1146/annurevmarine-042121-012329
- Bortolotti, L. E., St. Louis, V. L., & Vinebrooke, R. D. (2019). Drivers of ecosystem metabolism in restored and natural prairie wetlands. Canadian Journal of Fisheries and Aquatic Sciences, 76(12), 2396–2407. https://doi.org/10.1139/cjfas-2018-0419
- Brothers, S., Kazanjian, G., Köhler, J., Scharfenberger, U., & Hilt, S. (2017). Convective mixing and high littoral production established systematic errors in the diel oxygen curves of a shallow, eutrophic lake. *Limnology and Oceanography: Methods*, 15(5), 429–435. https://doi.org/ 10.1002/lom3.10169
- Brothers, S. M., Hilt, S., Meyer, S., & Köhler, J. (2013). Plant community structure determines primary productivity in shallow, eutrophic lakes. *Freshwater Biology*, 58(11), 2264–2276. https://doi.org/10.1111/fwb.12207
- Brylinsky, M., & Mann, K. H. (1973). An analysis of factors governing productivity in lakes and reservors. *Limnology & Oceanography*, 18(1), 1–14. https://doi.org/10.4319/10.1973.18.1.0001
- Burkholder, J. M., & Wetzel, R. G. (1989). Microbial colonization on natural and arificail macrophytes in a phosphorus-limited, Hardwater Lake. *Journal of Phycology*, 25(1), 55–65. https://doi.org/10.1111/j.0022-3646.1989.00055.x
- Cael, B. B., Heathcote, A. J., & Seekell, D. A. (2017). The volume and mean depth of Earth's lakes. *Geophysical Research Letters*, 44(1), 209–218. https://doi.org/10.1002/2016GL071378

- Cheng, F. Y., & Basu, N. B. (2017). Biogeochemical hotspots: Role of small water bodies in landscape nutrient processing. Water Resources Research, 53(6), 5038–5056. https://doi.org/10.1002/2016WR020102
- Cheng, F. Y., Park, J., Kumar, M., & Basu, N. B. (2023). Disconnectivity matters: The outsized role of small ephemeral wetlands in landscapescale nutrient retention. *Environmental Research Letters*, 18(2), 024018. https://doi.org/10.1088/1748-9326/acab17
- del Giorgio, P. A., & Williams, P. J. L. B. (2005). The global significance of respiration in aquatic ecosystems: From single cells to the biosphere. In P. delGiorgio & P. Williams (Eds.), *Respiration in aquatic ecosystems* (pp. 267–303). Oxford University Press. https://doi.org/10.1093/ acprof:oso/9780198527084.003.0014
- Downing, J. A. (2010). Emerging global role of small lakes and ponds: Little things mean a lot. *Limnética*, 29(1), 9–24. https://doi.org/10.23818/ limn.29.02
- Downing, J. A., Cole, J. J., Middelburg, J. J., Striegl, R. G., Duarte, C. M., Kortelainen, P., et al. (2008). Sediment organic carbon burial in agriculturally eutrophic impoundments over the last century. *Global Biogeochemical Cycles*, 22(1), GB1018. https://doi.org/10.1029/ 2006GB002854
- Duarte, C. M., & Kalff, J. (1989). The influence of catchment geology and lake depth on phytoplankton biomass. Archiv für Hydrobiologie, 115(1), 27–40. https://doi.org/10.1127/archiv-hydrobiol/115/1989/27
- Hagerthey, S. E., Cole, J. J., & Kilbane, D. (2010). Aquatic metabolism in the Everglades: Dominance of water column heterotrophy. *Limnology & Oceanography*, 55(2), 653–666. https://doi.org/10.4319/lo.2010.55.2.0653
- Hansen, A. T., Dolph, C. L., Foufoula-Georgiou, E., & Finlay, J. C. (2018). Contribution of wetlands to nitrate removal at the watershed scale. *Nature Geoscience*, 11(2), 127–132. https://doi.org/10.1038/s41561-017-0056-6
- Hanson, P. C., Bade, D. L., Carpenter, S. R., & Kratz, T. K. (2003). Lake metabolism: Relationships with dissolved organic carbon and phosphorus. *Limnology & Oceanography*, 48(3), 1112–1119. https://doi.org/10.4319/10.2003.48.3.1112
- Hoellein, T. J., Bruesewitz, D. A., & Richardson, D. C. (2013). Revisiting Odum (1956): A synthesis of aquatic ecosystem metabolism. *Limnology* & Oceanography, 58(6), 2089–2100. https://doi.org/10.4319/lo.2013.58.6.2089
- Holgerson, M. A. (2015). Drivers of carbon dioxide and methane supersaturation in small, temporary ponds. *Biogeochemistry*, 124(1–3), 305–318. https://doi.org/10.1007/s10533-015-0099-y
- Holgerson, M. A., Farr, E. R., & Raymond, P. A. (2017). Gas transfer velocities in small forested ponds. Journal of Geophysical Research: Biogeosciences, 122(5), 1011–1021. https://doi.org/10.1002/2016JG003734
- Holgerson, M. A., Hovel, R. A., Kelly, P. T., Bortolotti, L. E., Brentrup, J. A., Bellamy, A. R., et al. (2022b). Integrating ecosystem metabolism and consumer allochthony reveals nonlinear drivers in lake organic matter processing. *Linnology & Oceanography*, 67(S1). https://doi.org/10. 1002/Ino.11907
- Holgerson, M. A., Post, D. M., & Skelly, D. K. (2016). Reconciling the role of terrestrial leaves in pond food webs: A whole-ecosystem experiment. *Ecology*, 97(7), 1771–1782. https://doi.org/10.1890/15-1848.1
- Holgerson, M. A., & Raymond, P. A. (2016). Large contribution to inland water CO₂ and CH₄ emissions from very small ponds. *Nature Geoscience*, 9(3), 222–226. https://doi.org/10.1038/ngeo2654
- Holgerson, M. A., Richardson, D. C., Roith, J., Bortolotti, L. E., Finlay, K., Hornbach, D. J., et al. (2022). Classifying mixing regimes in ponds and shallow lakes. Water Resources Research, 58(7), e2022WR032522. https://doi.org/10.1029/2022WR032522
- Holgerson, M. A., Zappa, C. J., & Raymond, P. A. (2016). Substantial overnight reaeration by convective cooling discovered in pond ecosystems. *Geophysical Research Letters*, 43(15), 8044–8051. https://doi.org/10.1002/2016GL070206
- Hornbach, D. J., Hove, M. C., Ensley-Field, M. W., Glasenapp, M. R., Goodbar, I. A., Harman, J. D., et al. (2017). Comparison of ecosystem processes in a woodland and prairie pond with different hydroperiods. *Journal of Freshwater Ecology*, 32(1), 675–695. https://doi.org/10.1080/ 02705060.2017.1393468
- Hornbach, D. J., Schilling, E. G., & Kundel, H. (2020). Ecosystem metabolism in small ponds: The effects of floating-leaved macrophytes. *Water*, 12(5), 1458. https://doi.org/10.3390/w12051458
- Maranger, R., Jones, S. E., & Cotner, J. B. (2018). Stoichiometry of carbon, nitrogen, and phosphorus through the freshwater pipe. Limnology and Oceanography Letters, 3(3), 89–101. https://doi.org/10.1002/lol2.10080
- Martinsen, K. T., Andersen, M. R., Kragh, T., & Sand-Jensen, K. (2017). High rates and close diel coupling of primary production and ecosystem respiration in small, oligotrophic lakes. Aquatic Sciences, 79(4), 995–1007. https://doi.org/10.1007/s00027-017-0550-3
- Martinsen, K. T., Kragh, T., & Sand-Jensen, K. (2019). Carbon dioxide fluxes of air-exposed sediments and desiccating ponds. *Biogeochemistry*, 144(2), 165–180. https://doi.org/10.1007/s10533-019-00579-0
- Odum, H. T. (1956). Primary production in flowing waters. *Limnology & Oceanography*, 1(2), 102–117. https://doi.org/10.4319/10.1956.1.2.0102 Qin, B., Zhou, J., Elser, J. J., Gardner, W. S., Deng, J., & Brookes, J. D. (2020). Water depth underpins the relative roles and fates of nitrogen and
- phosphorus in lakes. Environmental Science & Technology, 54(6), 3191–3198. https://doi.org/10.1021/acs.est.9b05858 Rabaey, J., & Cotner, J. (2022). Pond greenhouse gas emissions controlled by duckweed coverage. Frontiers in Environmental Science, 10,
- 889289. https://doi.org/10.3389/fenvs.2022.889289 Rabaey, J. S., Domine, L. M., Zimmer, K. D., & Cotner, J. B. (2021). Winter oxygen regimes in clear and turbid shallow lakes. *Journal of*
- Geophysical Research: Biogeosciences, 126(3), e2020JG006065. https://doi.org/10.1029/2020JG006065 Rabaey, J. S., Holgerson, M., Richardson, D., Andersen, M., Bansal, S., Bortolotti, L., et al. (2024). Dataset for: Freshwater biogeochemical hotspots: High primary production and ecosystem respiration in shallow waterbodies ver 2 [Dataset]. *Environmental Data Initiative*. https://doi. org/10.6073/pasta/bf8105024133daa4bd41e791986156e3
- Richardson, D. C., Carey, C. C., Bruesewitz, D. A., & Weathers, K. C. (2017). Intra- and inter-annual variability in metabolism in an oligotrophic lake. Aquatic Sciences, 79(2), 319–333. https://doi.org/10.1007/s00027-016-0499-7
- Richardson, D. C., Holgerson, M. A., Farragher, M. J., Hoffman, K. K., King, K. B. S., Alfonso, M. B., et al. (2022). A functional definition to distinguish ponds from lakes and wetlands. *Scientific Reports*, 12(1), 10472. https://doi.org/10.1038/s41598-022-14569-0
- Rubbo, M. J., Cole, J. J., & Kiesecker, J. M. (2006). Terrestrial subsidies of organic carbon support net ecosystem production in temporary forest ponds: Evidence from an ecosystem experiment. *Ecosystems*, 9(7), 1170–1176. https://doi.org/10.1007/s10021-005-0009-6
- Sand-Jensen, K., Andersen, M. R., Martinsen, K. T., Borum, J., Kristensen, E., & Kragh, T. (2019). Shallow plant-dominated lakes Extreme environmental variability, carbon cycling and ecological species challenges. *Annals of Botany*, 124(3), 355–366. https://doi.org/10.1093/aob/ mcz084
- Schmadel, N. M., Harvey, J. W., Schwarz, G. E., Alexander, R. B., Gomez-Velez, J. D., Scott, D., & Ator, S. W. (2019). Small ponds in headwater catchments are a dominant influence on regional nutrient and sediment budgets. *Geophysical Research Letters*, 46(16), 9669–9677. https://doi. org/10.1029/2019GL083937

- Solomon, C. T., Bruesewitz, D. A., Richardson, D. C., Rose, K. C., Van de Bogert, M. C., Hanson, P. C., et al. (2013). Ecosystem respiration: Drivers of daily variability and background respiration in lakes around the globe. *Limnology & Oceanography*, 58(3), 849–866. https://doi.org/ 10.4319/lo.2013.58.3.0849
- Staehr, P. A., Baastrup-Spohr, L., Sand-Jensen, K., & Stedmon, C. (2012). Lake metabolism scales with lake morphometry and catchment conditions. Aquatic Sciences, 74(1), 155–169. https://doi.org/10.1007/s00027-011-0207-6
- Therneau, T., & Atkinson, B. (2022). rpart: Recursive partitioning and regression trees. Retrieved from https://CRAN.R-project.org/ package=rpart
- Theus, M. E., Ray, N. E., Bansal, S., & Holgerson, M. A. (2023). Submersed Macrophyte density regulates aquatic greenhouse gas emissions. Journal of Geophysical Research: Biogeosciences, 128(10), e2023JG007758. https://doi.org/10.1029/2023JG007758
- Vachon, D., & Prairie, Y. T. (2013). The ecosystem size and shape dependence of gas transfer velocity versus wind speed relationships in lakes. Canadian Journal of Fisheries and Aquatic Sciences, 70(12), 1757–1764. https://doi.org/10.1139/cjfas-2013-0241
- Vachon, D., Sadro, S., Bogard, M. J., Lapierre, J., Baulch, H. M., Rusak, J. A., et al. (2020). Paired O₂ –CO₂ measurements provide emergent insights into aquatic ecosystem function. *Limnology and Oceanography Letters*, 5(4), 287–294. https://doi.org/10.1002/lol2.10135
- Wetzel, R. G. (2001). Limnology: Lake and river ecosystems (3rd ed.). Academic Press.
- Winslow, L. A., Zwart, J. A., Batt, R. D., Dugan, H. A., Woolway, R. I., Corman, J. R., et al. (2016). LakeMetabolizer: An R package for estimating lake metabolism from free-water oxygen using diverse statistical models. *Inland Waters*, 6(4), 622–636. https://doi.org/10.1080/IW-6.4.883

References From the Supporting Information

- Hanson, P. C., Carpenter, S. R., Kimura, N., Wu, C., Cornelius, S. P., & Kratz, T. K. (2008). Evaluation of metabolism models for free-water dissolved oxygen methods in lakes: Free-water lake metabolism models. *Limnology and Oceanography: Methods*, 6(9), 454–465. https:// doi.org/10.4319/lom.2008.6.454
- O'Hara-Wild, M., Hyndman, R., & Wang, E. (2022). fable: Forecasting models for tidy time series. *R package version 0.3.2*. https://CRAN.R-project.org/package=fable